Author’s response to reviews

Title: Feature Selection through Validation and Un-censoring of Endovascular repair Survival Data for Predicting the Risk of Re-intervention

Authors:

Omneya Attallah (o.attallah@aast.edu)
Xianghong Ma (x.ma@aston.ac.uk)
Alan Karthikesalingam (alankarthi@gmail.com)
Peter Holt (peteholt@btinternet.com)
Matthew Thompson (matt.thompson@nhs.net)
Rob Sayers (rs152@leicester.ac.uk)
Matthew Bown (mjb42@leicester.ac.uk)
Eddie Choke (tcchoke@hotmail.com)

Version: 1 Date: 02 Aug 2016

Author’s response to reviews:

1. Confirm the ethics approval of the study.

I added a paragraph in declaration section page 29.

"The study (including anonymised imaging data from Leicester and St George's hospital patients) was classified as service evaluation by communication with Wandsworth Research Ethics Committee REC in 2012, when the data analysis plan was conceived. Wandsworth REC waived the need for approval as this was a retrospective observational study of anonymised data."

2. Comment 2

I change the Methodology section to Methods.

3. Comment 3
I shortened the conclusion section and moved some parts to discussion section lines 519-613.

4. Comment 4
I put a declaration section with the following subsection
- Ethics approval and consent to participate
- Consent to publish
- Availability of data and materials
- Competing interests
- Funding
- Authors' Contributions
- Acknowledgements
- Authors' Information is in the cover page

5. Comment 5
I selected 'Standards, technology, and modeling' during resubmission.